



The Role of the Circular Economy in Road Transport to Mitigate Climate Change and Reduce Resource Depletion

Victor Hugo Souza De Abreu, Mariane Gonzalez Da Costa, Valeria Xavier Da Costa, Tassia Faria De Assis, Andrea Souza Santos *[®] and Marcio de Almeida D'Agosto

Program of Transportation Engineering, Alberto Luiz Coimbra Institute for Graduate Studies and Research in Engineering (COPPE), Technology Center, Federal University of Rio de Janeiro (UFRJ), Rio de Janeiro 21941-914, Brazil; victor@pet.coppe.ufrj.br (V.H.S.D.A.); mariane.gonzalez@pet.coppe.ufrj.br (M.G.D.C.); vxavier@pet.coppe.ufrj.br (V.X.D.C.); tassiafa@pet.coppe.ufrj.br (T.F.D.A.); dagosto@pet.coppe.ufrj.br (M.d.A.D.) * Correspondence: and real sources of the source uffet br

* Correspondence: andrea.santos@pet.coppe.ufrj.br

Abstract: The transport sector is responsible for several environmental impacts, including contributions to climate change through greenhouse gas emissions and depleting natural resources. A strategy to reduce these issues goes towards the application of a circular economy, a concept that offers a response to increasing concerns about resource scarcity and the associated impacts from their use. Thus, this paper aims to fill a gap in the literature that consists of the scarcity of studies that consider the circular economy application on a micro, meso, and macro level in road transport, including all stages as well as the 7 Rs of the reverse cycle. Therefore, an approach is presented to meet road transport needs, highlighting best practices obtained through a literature review, to promote climate change mitigation and resource depletion. Qualitative data were presented for each circular economy stage with 46 best practices identified, providing invaluable guidance to transport decision-makers. Thus, public policies focusing on all of the CE stages should be taken into consideration, not only those responsible for closing the cycle, such as waste and recycling or disposal and treatment.

Keywords: circular economy; sustainable development; road transport; best practices; climate change mitigation; resource depletion

1. Introduction

With climate change affecting all of world's regions and with resource scarcity exacerbated by global population, urbanization, and economic growth, a transition that prioritizes sustainability is recognized as critical and urgent [1,2]. In this context, the 2015 Paris Agreement "aims to strengthen the global response to climate change threat with efforts to limit global average temperature to below 2 °C, preferably to 1.5 °C above preindustrial levels" [3], where greenhouse gas (GHG) emissions linked to anthropic causes must decrease 40–50% by 2050 to prevent warming exceeding 2 °C [4]. Recently, during the UN Climate Change Conference in Glasgow (COP26) in 2021, it was identified that although there is a collective political will to meet the goals of the Paris Agreement, the current climate plans implemented by member countries are still far from meeting the Nationally Determined Contributions (NDCs) [5,6]. The Glasgow Climate Pact calls for all countries to submit stronger national action plans by the year 2022, instead of 2025, which was the original schedule.

Thus, the GHG emissions from the transport sector is a relevant factor to be considered. While important for urban mobility, road transport is also a major user of carbon-based fuels [7]. Over 60% of all oil consumed in the world goes to the transport sector, of which 76% is consumed by road travel [8]. This is especially true in an urban context since 15% of the total energy worldwide is used for urban passenger and freight transport [9]. This



Citation: De Abreu, V.H.S.; Da Costa, M.G.; Da Costa, V.X.; De Assis, T.F.; Santos, A.S.; D'Agosto, M.d.A. The Role of the Circular Economy in Road Transport to Mitigate Climate Change and Reduce Resource Depletion. *Sustainability* **2022**, *14*, 8951. https:// doi.org/10.3390/su14148951

Academic Editors: J. Augusto Felício and Vitor Caldeirinha

Received: 7 May 2022 Accepted: 19 July 2022 Published: 21 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). reality may become even worse because it is estimated that the urban global population will reach 68% by 2050 [10].

In terms of direct CO₂ emissions from fuel combustion, transport is responsible for 24%, with road vehicles accounting for nearly three-quarters of this total [11]. The transport sector currently emits about 9.7 GtCO₂eq, and if no mitigation policies are implemented, transport emissions could reach around 10 to 18 GtCO₂eq in 2050 [12]. Changes in transport, such as efficiency improvements versus maintaining the business-as-usual scenario, is a choice that will influence communities, regional economies, and other variables [7].

A response that should reduce the adverse impact on the natural environment is found in CE principles' employment, which can be a path to change in the contemporary system of production and consumption [13]. CE principles comprise means to support both economic growth and sustainable resource management and offer a positive opportunity for change [14], due to the increasing concern for sustainability [15] and linked environmental impacts, such as resource waste [16] and GHG emissions [13].

Acceptance by national and subnational governments of the Paris Agreement framework together with circular economy (CE) strategies are essential to understanding its potential effect on climate change mitigation [17]. In terms of the necessity of increasing ambition to reduce GHG emissions, some countries have announced phasing out sales of new internal combustion engine vehicles. As well, the number of countries announcing pledges to achieve net-zero emissions by 2050 is growing [18].

This study undertakes a literature review of the recent CE concept applied to transportation in view of providing consolidated information to support strategies that assist transport decision makers to adopt best practices and minimize the impacts of the road transport sector on sustainable development. In particular, the review focuses on GHG emissions, maintaining products and materials in use, regenerating natural systems, representing a systemic change that builds long-term resilience, and providing environmental and social benefits. Haddach et al. [19] conceptualize best practices as processes, techniques, technological innovations, equipment development, or resources capable of promoting the relevant enhancement of quality, cost, safety, performance, environment, and any other concept that impacts society.

It is worth noting that new technologies, such as electric or hydrogen-powered vehicles are used as best practices for promoting CE stages considering the road transport sector in a general context; however, a specific analysis for each of these vehicle types is not performed and this is a limitation of the study.

Subsided by the literature review, this paper presents the CE stages applied to transport, the life cycle stages of vehicle production and operation, as well as energy consumption and a framework of the best practices existent in literature to be applied at micro, meso, and macro levels in a transportation system. The paper also aligns all CE stages to the 2030 Agenda for Sustainable Development and its 17 Sustainable Development Goals (SDGs) [20], within which this study is mainly related to "affordable and clean energy" (Goal 7), "sustainable cities and communities" (Goal 11), "responsible consumption and production" (Goal 12), and "climate action" (Goal 13). The relevance of this study is in addressing a subject of global amplitude, which has been the subject of debate in reports such as the IPCC (2022), which still lacks related research, especially regarding the CE stages that, although not responsible for closing the cycle, are of great importance in reducing GHG emissions and the depletion of natural resources.

To fulfill our objectives, in addition to this introductory section: Section 2 presents a conceptualization of CE, highlighting its advances concerning linear economy; Section 3 shows how the methodological process was carried out to effectively conduct the literature review, pointing out the main gaps identified in knowledge; Section 4 presents a complete CE cycle proposed by authors and also highlights the best CE practices identified in the literature that can be implemented by transport decision makers, as well as the 7 Rs of reverse logistics; Section 5 describes in detail each of the best practices outlined in Section 4;

3 of 26

and finally, Section 6 presents the final considerations, which also contain proposals for future studies.

2. Theoretical Backgrounds

In this section, the main concepts of CE are presented utilizing a comparison with the linear economy. In addition, the basic CE stages considered in this study are also defined. Both scholars and practitioners claim that consideration of CE is useful for spurring sustainable development [21], thus "it is imperative to provide a comprehensive review of the CE concept" [1] which has been introduced as "an economic model for the transition from the linear to the circular system" [22]. For many years, the description of the linear economy (Figure 1) as "take, make, waste" has been broadly recognized, but nowadays, a new concept has emerged [23].



Figure 1. Steps of literature review.

In general, the linear economy's constraints are overcome by CE since it promotes cyclical flows of resources in the production-consumption system [24], thus moving towards sustainability [25]. The circular model is an economic system that promotes maintenance activities, sharing, leasing, reuse, remanufacturing, and recycling within closed loops to promote sustainability in management throughout the system's life cycle [26].

CE aims to reproduce the circularity of ecosystems "by transforming wastes into resources, establishing a closed material cycle through minimization of dissipative emissions, dematerializing products and economic activities and decarbonizing energy" [24]. Ghisellini et al. [27] found in the literature that the CE concept root originated from ecological and environmental economics and industrial ecology. Different meanings have been used to describe CE with the concept of a cyclical closed-loop system as a common point [28].

In short, CE is a holistic system that serves a waste-free production-consumption system in such a way as to reduce the consumption of natural resources and energy. CE is a concept of creating value through the rational use of resources to minimize the adverse environmental impact of our societies at all phases of a product life cycle, which enables the used products to be collected and reused, recycled, or remanufactured [13].

The CE can be applied at different levels: (i) micro: single process; (ii) meso: ecoindustrial parks; and (iii) macro: local, regional, and national economies [27,28]. The micro level is related to CE implementation by a company of cleaner production and industrial ecology [29] and ecodesign [27] practices. The meso level regards the network between companies leading to industrial symbiosis [28,29]. The application at the macro level regards legislation to promote zero waste programs, eco-cities, and collaborative consumption [27].

3. Methodology and Gap Analysis

This section describes the procedure used to conduct the literature review, focusing on relevant and current studies on the subject, as well as highlighting the main knowledge gaps that the study intends to address. A review of transport-related best practices for each CE stage was completed, prioritizing international studies and reports selected from the literature, as shown in Figure 1.

The literature search was performed in Web of Science and Scopus databases by using a search guided by topics "road transport*" and "circular economy". Additionally, documentary research was conducted as a complementary source for gathering best practices, wherein reports and guidelines were considered. It is worth noting that the choice of keywords and their combinations considered a brainstorming process to choose the most relevant keywords. Subsequently, a team composed of academics and professionals in the transport field refined these keywords to provide solid validity.

However, since initial research in databases provided a very limited number of relevant studies that were directly applicable to transport and that considered vehicle parts and components, operation, and emissions, one increased the research field by using other relevant terms to road transport such as "vehicle", "automotive industry" and "supply chain". Figure 2 illustrates the intersection between the main themes of the article and guided the authors to identify the keywords used and their combination using Boolean logic operators.



Figure 2. Keywords used in the queries for the literature search.

Thus, 42 studies were identified with the quality and applicability necessary to be incorporated in the research repository, consisting of 40 articles and 2 reports that address CE application or, at least, the closed-loop concept in road transport, which assists in mitigating climate change and reducing resource depletion. Although the focus of this paper is not to carry out bibliometric approaches to research repositories, it is worth noting that a gradual evolution of the topic was perceived, mainly from 2015 on, with a peak in the number of publications in 2019, which accounts for 26% of the studies; however, papers in 2022 are still being published. In addition, studies based on the closed-loop concept were widely used in the 90s.

The fact that 80% of studies were published in the last four years points to the relevance of the topic and indicates that it will be fruitful for new research. Additionally, these studies were published in important periodicals, such as the Journal of Cleaner Production, which accounts for 24% of publications considered here.

The bibliometric analysis also indicates investigations in several areas that seek to identify how to incorporate CE concepts in road transport (and are published in relevant periodicals) in topics such as waste management (Waste Management, Resources, Conservation & Recycling, and Journal of Material Cycles and Waste Management), energy (Energy, Journal of Energy Storage, and Energy Reports), and environmental science (Journal of Environmental Sciences, Sustainability, and Journal of Environmental Management).

Nonetheless, it was also found that transport as the main subject was considered by only around 62% of the studies. Most of the studies are concentrated mainly on the last CE stages: waste and its disposal and treatment, and recycling as the most considered of the 7 Rs, as shown in Figure 3.





However, although the best practices to achieve a circularity in the last CE stages are relevant, others with the same purpose can be found in studies wherein other CE stages are considered, e.g., best practices applied to manufacturing can facilitate future remanufacturing. The application of best practices at each cycle stage can help mitigate the impacts of climate change and resource depletion in the entire life cycle of the vehicle. Although these actions have a more specific application, they focus on one or two cycle stages, promoting a systemic benefit.

In addition, with a view of the identification of best practices that serve one or more stages of the cycle, other studies were selected that, despite not specifically addressing the CE concept, at least consider the closed-loop concept. This strategy was adopted specifically to fulfill the objectives of our research, which, differently from current literature, seeks to include all CE stages applied to road transport.

However, reduction, reuse and repair, and emissions stages are subjects that need more investigation considering the CE for road transportation and how that impacts climate change. It is noteworthy that the non-consideration of all stages does not concern just one application stage but the lack of an overview of the whole life cycle for transport required to achieve circularity and the impact of each CE stage on climate change. In addition, only approximately 62% of studies consider transport as the main subject. Other publications identified in complementary research indicate neither the application of the CE concept specifically to the transport life cycle, nor to the vehicle, fuel, or energy. The analyses of these studies are mostly focused on waste management and reverse logistics, wherein transport is only being used as a tool to optimize this process. This approach does not consider energy use, which in the context of transport operations does not generate waste, but GHG emissions and pollutants.

Policies used to promote best practices related to CE stages are another issue that should be considered. Saidani et al. [30] identified a lack of regulatory frameworks in the U.S., wherein neither quantitative targets for recycling light or heavy vehicles nor national regulations were created. Meanwhile, in Europe, the legislation is more consolidated and seeks to encourage the reuse of automotive parts, as well as the remanufacturing and recycling market associated with them. Nevertheless, policies are a key action lever to facilitate the entry of end-of-life vehicle components into appropriate circular loops.

Another obstacle is the difficulty in accessing standardized data by actors involved in the circularity process of vehicle parts at their end of life. The access to information on the extraction of raw material, original vehicle parts, component production [30], and data transparency are essential to facilitating the replacement with recycled parts demanded by the automotive recycling market at appropriate prices [29,30]. Policies related to air quality and climate change have been used to reduce pollutants and GHG emissions through the imposition of limits on regulated emissions from fossil fuel combustion.

4. Results

According to the literature review, the CE concept for transport could be divided into a first and second life cycle and should consider the final disposal for residues that cannot be recycled, as expressed in Figure 4. The first cycle considers eight stages: (1) raw materials; (2) ecodesign; (3) manufacturing; (4) distribution; (5) using and consuming; (6) collection; (7) waste or emissions; and (8) disposal and treatment. The reverse cycle is divided into the 7 Rs, which can be considered at more than one cycle stage: (1) reduce; (2) refurbish; (3) remanufacture; (4) redistribute; (5) reuse; (6) repair; and (7) recycle.



Figure 4. Circular economy stages applied to transport.

Table 1 provides an overview for each CE stage that will be further discussed in Section 4.

Stages	Description		
Raw Materials	Raw materials are inputs usually extracted or produced from natural resources or obtained through the recovery or recycling of solid waste before final disposal [31], which allow activities such as transport to continue operating, mainly using energy [32].		
Ecodesign	The first concept in ecodesign was established as the improvement of environmental aspects of the performance of the product throughout its lin cycle, according to the Ecodesign Work Plan [33]. The product is thus designed to reduce the use of raw material and is created to last through i design life cycle and then be available for eventual reuse or recycling.		
Manufacturing	Manufacturing is a process that can be sustainable if the concept of cleaner production is applied. This concept was defined by the United Nations Environment Programme (UNEP) as an integrated environmental strategy applied to the production process to increase ecoefficiency and reduce environmental impact [34].		
Distribution	The distribution segment of the chain is positioned between the main manufacturer and final consumer, where it is more common to transport finished or partially finished goods in transfers between factory warehouses and their distribution centers or wholesalers' warehouses, which will distribute them to end customers [35].		
Using and Consuming	Using and Consuming The consumption process, together with production/distribution and vehicle use, supplement the concept of "end of life" by reducing, redefining, recycling, and recovering materials [36].		
Collection	Collection and transport is a critical process of waste management [23], especially when considered together with concerns about social and environmental impacts related to emissions of pollutants and implications for the worsening of urban air quality due to fossil fuel consumption [9].		
Disposal and Treatment	The treatments of a product are mechanisms that can help reduce the volume and toxicity of waste before disposal, and can be classified as biological, physical, and chemical [37]		

 Table 1. Description of each CE stage considered in the research.

Complementarily, a brief description of each of the 7 Rs of the reverse cycle is presented in Table 2.

Table 2. Description of each 7 Rs of the reverse cycle considered in the research.

Reverse Cycle	Description	
Reduce	Reduction is the process of decreasing the exploitation of natural resources and energy consumption [13].	
Refurbish	Refurbishing is the process of returning a product to good working condition, replacing or repairing major components that are defective or close to failure [38].	
Remanufacture	Remanufacturing comprises activities to extend the life of the product through repairs, restoration, and upgrading [25].	
Redistribute	Redistribution comprises the reverse logistics [39].	
Reuse	"Any operation by which products or components that are not waste are used again for the same purpose for which were conceived" [40]	
Repair	Repair consists of restoring existing items [38].	
Recycle	Recycling is defined by product reprocessing and secondary material recovery activities used to manufacture new products [25].	

The proposed overview of all CE stages aim to consider more than the closed loop stages' recycling strategy of the waste generated in the product's closed life cycle flux

(cradle-to-cradle), that is, further studies should also give special attention to the application of best CE practices in other life cycle stages that are poorly investigated in the scientific literature, such as raw materials, ecodesign, and using and consuming, as already discussed in Section 3, especially when it comes to transport, which considers both energy consumption and the vehicle's life cycle.

The vehicle as a product entirely fits into the concepts of CE. On the other hand, the energy consumed by a vehicle, whether through fuel combustion or electricity, emits GHG and pollutants into the atmosphere that fit into the concept of CE, which also considers energy loop and emissions [27–30,41–43]. The CE concept has gained force due its connection to other strategies such as cradle-to-cradle design [29] and the life cycle approach. Figure 5 shows the life cycle stages of a vehicle's production and operation, as well as energy consumption, and a framework is presented that considers interactions between the well-to-wheel, cradle-to-gate, and the complete life cycle (cradle-to-cradle).



Figure 5. Life cycle stages of the transportation system, considering the flow for the production, operation, and EoL, as well as the energy consumption. Source: Bauer et al. [44].

The well-to-wheel LCA approach refers to the entire flow of energy consumed by vehicle operation and emitted to the environment in the form of GHG, from extraction and transportation to consumption in the vehicle operation stage. The cradle-to-gate approach refers to the production stage of the vehicle, considering the extraction of raw materials, production parts, and vehicle manufacturing, as well as the transportation of resources, parts, and components, until reaching the gate of the industry.

The cradle-to-grave approach is a wider concept, comprising the cradle-to-gate approach together with the distribution of the vehicle to consumers, the maintenance while it is in operation, and the EoL, then contemplating the waste management, and when necessary, their treatment and correct final disposal in landfills. It is worth noting that the 7 Rs are applied, in majority, at the EoL to ensure circularity.

The best practices found in the literature for each stage and the implementation level, as described in Section 2, are highlighted in Table 3 to assist decision makers in developing strategies that apply to all stages of the life cycle of road transportation and minimize

GHG emissions. The results of the application of the LCA tool allowed us to identify opportunities to improve environmental performance [45] in each stage, wherein CE best practices could be implemented to achieve circularity.

 Table 3. CE-related best practices and their implementation level.

CE Stages	Best Practices	Implementation Level	Life Cycle
Raw Materials/Reduce	Utilization of secondary resource	micro	cradle-to-cradle (7 Rs)
	Investing in recycling technologies	micro	cradle-to-cradle (7 Rs)
	Increasing the use of recycled material	micro	cradle-to-cradle (7 Rs)
	Re-use of product such as second life of vehicle batteries in residential sector	macro	cradle-to-cradle (7 Rs)
	Look for new technologies for use of waste or coproducts as raw material	meso	cradle-to-cradle (7 Rs)
	Develop low aerodynamic resistance vehicle	micro	cradle-to-gate
	Use less raw material by rethinking the vehicle design	micro	cradle-to-gate
Refurbish	Rethink propulsion system toward renewable and alternative energy use	micro	cradle-to-gate
	Choose materials that require energy efficiency in manufacturing processes	micro	cradle-to-gate
ign/	Think more durable and better-functioning products	micro	cradle-to-gate
odes	Modularity that allows interchangeable parts	micro	cradle-to-gate
fanufacturing/ Remanufacture	Refurbish products to reincorporate them to the economic system, turning more attractive by updating it appearance	micro	cradle-to-cradle (7 Rs)
	Rethink the vehicle to use lighter material in manufacturing the vehicle to reduce fuel consumption in the operation	micro	cradle-to-gate
	Less energy and material consumption in production process	micro	cradle-to-gate
	Improve energy-efficiency in manufacturing	micro	cradle-to-gate
	Vehicles with materials that will facilitate reusing, recycling and dismantling	micro/macro	cradle-to-gate
4 7	Policies to promote remanufactures	macro	cradle-to-cradle (7 Rs)
	Implementation of distribution and freight consolidation centers	meso	cradle-to-gate and well-to-wheel
on ion/	Night-time collection and distribution and route optimization	micro	Operation (end use)
buti rtat tribı	Use of cleaner mode of transport and energy efficient logistics	micro	Operation (end use)
Distril Transpo Redist	Reduce necessity of delivery services by using new technologies, e.g., 3D printers	micro	cradle-to-gate
	Routing of a local delivery system configuration and use of information systems	micro	Operation (end use)
Use and consumption/ Reuse	Adoption of low Emission Zones in urban areas	macro	Operation (end use)
	Toll to fight traffic congestion	macro	Operation (end use)
	Sharing economy and collaborative consumption	meso	Operation (end use)
	Vehicle occupancy optimization	micro	Operation (end use)
	Use of different types of vehicles	micro	Operation (end use)

CE Stages	Best Practices	Implementation Level	Life Cycle
Collection Transportation/ Recollect	Optimization of the number of vehicles and containers	micro	Operation (end use)
	Route selection and optimization	micro	Operation (end use)
	Limited disposal in landfill	micro	Operation (end use)
	Techniques for computational optimization of transfer station	micro	Operation (end use)
	Digital connection mechanism with consumers	micro	Operation (end use)
Waste/Recycling and Emission	Increased recycling of vehicle components	micro/macro	cradle-to-cradle (7 Rs)
	Promoting emerging technologies for propulsion of vehicles with zero emission	Micro/macro	Cradle-to-gate
	Use of electric vehicles with fewer parts and components unlike the combustion engine	micro/macro	cradle-to-cradle (7 Rs)
	Establish policies and regulation for end-of-life vehicles (ELVs) and generating energy	macro	Cradle-to-grave
	Method to optimization of municipal solid waste treatment	micro/macro	Operation (end use)
	Renewable Transport Fuel Obligation (RTFO) alike laws	macro	Well-to-wheel
Disposal and Treatment	Guidelines and regulations for the disposal of EVLs and oil residues	macro	cradle-to-cradle (7 Rs)
	Standardization in procedure of dismantling and technologies	macro	cradle-to-cradle (7 Rs)
	ASR must be properly characterized before recovered	micro/meso	cradle-to-cradle (7 Rs)
	Energy recovery from the heat created at the incineration of combustible waste	micro	cradle-to-cradle (7 Rs)
	Techniques to minimize the costs of waste treatment, e.g., the use of computational optimization	micro	Operation (end use)
	Targets for reuse, recycling and recovery of EVLs	micro/meso/macro	cradle-to-cradle (7 Rs)
	Encouraging new vehicle design without heavy metals	macro	Cradle-to-gate
	Incineration as an alternative to reduce the volume of waste	macro/macro	cradle-to-cradle (7 Rs)

Table 3. Cont.

5. Discussion

It worth noting that some best practices adopted in one stage of CE can impact another, especially when it concerns the reverse flux, e.g., recycling batteries can reduce the need for raw materials extraction. Therefore, the CE framework proposed for the road transport system and the concept and *status quo* of each step are discussed considering the direct and inverse related flux.

5.1. Application of Circular Economy Stages to Transport

This section discusses the best practices identified in the literature that can be applied at each CE stage as well as the 7 Rs that reduce the impacts of the road transport sector on climate change and natural resource depletion.

5.1.1. Ecodesign and Refurbish

The concept of ecodesign can be extended from product development to design environments, logistic chains, or services [46]. Therefore, ecodesign itself can be considered a best practice. Ecodesign reduces resource consumption, promotes the use of recyclable materials, and avoids the use of hazardous materials, being a best practice of sustainable supply chain management [47].

To Al-Sheyadi et al. [46], ecodesign consists of generating more durable products or products and production processes less intensive in energy use, considering it as a best

practice of green supply chain management. The changes in product design can be oriented toward the use of less raw material, or of energy sources that minimize emissions [48] and pollutants by designing the product to be easily disassembled, remanufactured, or recycled [46].

In that way, ecodesign considers the choice of materials that require energy efficiency in manufacturing processes; more durable and better-functioning products generate less waste, regulating the emissions of waste of the process system, and modularity allows the use of interchangeable parts that can be replaced in case of a defect, thus avoiding the exchange of the entire product, which also generates less waste. It also includes the design of products that are able to survive their life cycle and be reused for other functions after their first use.

Another best practice is refurbishing products and returning them to market in good condition, extending their life by replacing or repairing their faulty components. It is also recommended that their appearance be updated [38]. An example of best practices for transport considering ecodesign is to structure the vehicle to provide low aerodynamic resistance to decrease fuel consumption, or to rethink the vehicle propulsion system with a view of renewable and alternative energy use [49].

5.1.2. Raw Materials and Reduce

Urban sustainability is a model that contributes to improving people's quality of life, which is causally related to the needs and limits of natural resources [50]. The main challenges of this stage are minimizing the use of materials and increasing the reuse of natural resources [51].

The main raw materials required for vehicle production are: (i) metals, such as steel, aluminum, and copper, mainly responsible for composition of the body, chassis, and engine parts; (ii) glass; (iii) rubber, found mainly in tires; (iv) special fibers; and (vii) materials comprising other parts, such as batteries [52]. In the case of energy resources, the main sources of fossil energy are diesel and gasoline, and the main alternative and transition energy sources are electricity, gaseous fuels (natural gas, hydrogen, and liquefied petroleum gas), biofuels, such as alcohols (methanol and ethanol) and biodiesel, and others [53].

Important sources of secondary raw materials can be found in ELVs [54]. The use of this secondary resource, when aligned with the investment in recycling technologies and the increased use of recycled material, provides a promising prospect [55] and follows the SDG, which promotes flexible and resistant infrastructure, inclusive and sustainable industrialization, and the nurturing of innovation.

However, concerning the economic aspect, the costs of end-of-life operations—applying best available techniques to preserve high-quality recycling—are expected to become higher over the coming years, especially due to new materials and equipment required in vehicles and the vast expansion of the electric vehicle (EV) market [26].

According to Cusenza et al. [56], new ideas are being inspired by the construction and transport sectors for the reuse of EV batteries in residential areas, along with renewable energy generation technologies to match the highly renewable forms of electricity generation with demand.

In view of respecting the biogeochemical cycles concept that should be implicit in the CE [28], the rate of the removal flux of resources and their release into the environment should be minimized to become closer to the natural levels. In this line of thought, distinguishing renewable and non-renewable sources allows for the determination of the rate of extraction and emission toward the capacity for environment regeneration [29]. In the case of renewable energy, the rate of extraction must be lower than the capacity of absorption by nature. On the other hand, non-renewable energy exploitation should be lower than the rate of creation of new sources of clean and renewable sources [29].

Regarding energy use by vehicle operation, zero emission by energy transition goals aim to reduce fossil fuel consumption and seek new clean and renewable energy alternatives, such as the synthesis of liquid fuels from waste as a means of approaching the goal of zero landfilling [57] and closing the cycle, or the use of a coproduct of one process as a resource for biofuel production, e.g., the biodiesel production from the residual oil of the ethanol production process [58,59].

5.1.3. Manufacturing and Remanufacturing

A rethinking of product design and recycling are subsequently discussed. Changes in production processes are based on better process control, equipment modification, technology change, and changes in input materials [48]. Each of these changes can reduce energy consumption, improving the process's energy efficiency and consequently reducing emissions.

Among the techniques presented by El-Haggar [48], one highlights the practices that are aligned with the objective of this study, such as (1) changes in the production process to consume less energy and material, while also considering the reduction of losses and leakages; (2) rethinking the product design to consume less raw materials; and (3) recycling.

The industry was responsible for 24% of global emissions in 2018 [60]. Specifically, the automotive industry produced 9% of all annual global GHG emissions in the same year [61]. The automotive sector not only emits CO_2 directly from vehicle fuel combustion but also indirectly from vehicle manufacturing itself and the materials and parts supply chain [62].

Green manufacturing, as in the system proposed by Deif [63], is related to the ecofriendly production process wherein the environmental impacts are considered in the efficiency planning and control. Several studies pointed to the reduction of energy consumption as a best practice in manufacturing, i.e., making its production processes more energy efficient [64–67]. Practices to reduce electric energy consumption in manufacturing result in less GHG emissions, since the life cycle is considered, and emissions for electricity generation might be considered. A best practice that can be considered in this case is the use of clean energy, such as solar energy [66].

In addition, green manufacturing can also indirectly contribute to decreasing emissions of GHG. Specifically, this can be achieved by producing vehicles with materials that facilitate reusing, recycling, and dismantling [68]. Green manufacturing of vehicles can also contribute to employing a rethinking of materials used to produce lighter vehicles [49]. This contributes to more efficient fuel consumption and, consequently, reduced GHG emissions.

In addition to the aforementioned best practices, the remanufacturing process can also contribute to reducing climate change and reaching a country's SDGs [69]. According to the Ellen MacArthur Foundation [38], remanufacturing is "a process of disassembly and recovery at the subassembly or component level. Functioning, reusable parts are taken out of a used product and rebuilt into a new one". However, to achieve the needed demand at end-of-life resulting in increased vehicle sales, policies must be directed toward improving remanufacturing technologies and encouraging remanufacturing by the automotive industry [69].

The study developed by Saidani et al. [70] deals with the end-of-life of heavy vehicles and their second life since most of them are refurbished or remanufactured; for example, in the United Kingdom, about 50% are refurbished and 43% are remanufactured. Almost all the refurbished heavy vehicles are resold, mainly in developing countries. In Sweden, around 50% are resold after five years. However, the study has pointed to the disparities between developing and developed countries, since, even in Europe, remanufacturing activities are not completely organized; thus, in developing countries, resold vehicles at their end-of-life stage are probably not refurbished or remanufactured to extend their life.

5.1.4. Distribution Transport and Redistribute

Optimizing the system of distribution is necessary to promote CE and can be represented by modeling approaches for closing and slowing resource loops [71]. Ways to promote emission reduction include choosing cleaner transport modes, energy-efficient logistics, and do-it-yourself (DIY) devices (such as 3D printers), reducing the necessity for delivery services [72]. Throughout the material flow process, the use of refined distribution procedures to reduce transport requests and encourage recycling and reusing waste, e.g., through a closed loop, can further decrease waste and GHG emission [73].

As for distribution, the best practices reported in the literature, such as Barbosa et al. [74], involve such measures as a choice of facility locations, implementation of a distribution center and freight consolidation centers in urban areas, and distance reduction utilizing routing optimization, for example. They also cited the use of time windows and off-peak deliveries as strategic measures to reduce the impact generated by the distribution activity, such as night delivery, as well as an information system focused on problems such as tracking and monitoring the fleet.

Production return and distribution return comprise reverse logistics: production return involves the process by which firms might re-use, repair, or recover defective or superfluous products; distribution return can be represented by the removal from the market of an unsold product, or one not damaged during transport, as well as the redistribution from warehouses of products that can be renewed, such as pallets and packaging [75]. Therefore, reverse logistics is an important dimension of CE that enables the management of economic, social, and environmental challenges [71].

In addition, redistribution can include practices such as the reallocation of items or services no longer needed by some person or in some place [76]. This is considered together with collection, inspection, separation, reprocessing, and disposal as part of the recovery process category [77]. A redistribution system can be enhanced through a best practice, such as the reconfiguration of a local delivery system [78].

5.1.5. Use and Consumption and Reuse and/or Repair

The next set of studies outlined recommended practices for the use of urban vehicles in such ways as to reduce GHG emissions and air pollutants, exemplifying the consumption process:

- Low Emission Zones, prescribed by a law requiring drivers to have a special environmental sticker on their cars [79];
- A toll to combat traffic congestion implemented to discourage personal vehicle entry into urban centers and encourage people to use more efficient means, such as public transportation with increased availability [80];
- Use of the Friendly City concept to promote the development of comfortable living conditions and encourage citizens to stop using their cars, prioritizing a comfortable urban environment and a healthy city [81]; and
- Full or partial refund for the amount of electric energy used in the power supply to an employee using an electric or hybrid personal car, as well as taxation for the use of fossil fuel [80].

Based on Camacho-Otero et al. [36], collaborative consumption and sharing economy are ways to intensify the use of assets, valuing the use of idle resources rather than possession, and facilitating the reuse of products. The sharing economy can be conceptualized as a system where consumers give each other temporary access to the unused capacity of their goods, such as cars and houses, possibly for money [82].

Collaborative consumption is a series of resource circulation processes that help consumers obtain and supply, on a temporary or permanent basis, important resources or services through direct interaction with other clients or through an intermediary [83]. These actions happen in organized systems or networks, in which participants perform sharing activities in the form of rent, loans, and negotiation or the exchange of goods, services, transport solutions, space, or money [84].

The sharing economy is represented by activities such as: strengthening the use of durable goods, recirculating goods, sharing productive assets, and exchanging services [36]. Initiatives applied to transport include carpooling, car-sharing with or without a driver, and freight transport [85].

There is potential to reduce GHG emissions, as well as local air pollution, noise, traffic congestion, etc., due to the decrease in the use and production of private cars, since most of these initiatives are in cities whose problem with transport is greater. Among the environmental impacts caused by individual vehicles, production of automobiles through the extraction of raw materials; the transformation of materials and assembly of automobiles; replacement and production of spare parts, such as tires, batteries, lubricants, and coolants; fuel transformation processes that precede fuel consumption; and fuel consumption on the road are considered [85].

Other relevant best practices support reduced emissions, such as those indicated by Barbosa et al. [74]: (i) vehicle occupancy optimization, (ii) use of different types of vehicles to carry out deliveries and collections, (iii) shifting freight transport to cleaner modes (modal shift), (iv) use of vehicles with greater energy efficiency (fuel consumption), (v) use of alternative vehicles (propulsion systems), and (vi) reducing the weight of vehicles.

In addition, there are recommendations for the improvement of vehicle development: (i) the use of low rolling resistance tires, (ii) fleet renovation and modernization, (iii) promotion of improvements in vehicle aerodynamics, (iv) preventive maintenance of vehicles, (v) use of cleaner energy sources, and (vi) use of additives to improve the energy efficiency of fuels [86].

Additionally, integrated urban planning initiatives offer substantial opportunities for reducing the carbon footprint: a shift to public transportation can significantly reduce emissions; projects to upgrade bikeways and sidewalks can provide large emission reductions; the combination of rapid bus transit and walking offers potential for substantial mitigation of the CO_2 generated in urban transport, while lowering costs of mobility, and the adoption of a short-term and a long-term strategy for addressing carbon emissions are all important best practices [87].

5.1.6. Collection and Transport

End-of-life vehicle collection is a rapidly growing waste stream, so specific collection activities are required to avoid environmental pollution and resource depletion fed by the old linear economy model, "make-take-use-dispose" [88]. Thus, there is a variety of waste collection and transport methods aligned with CE [23]; improving the efficiency of this activity generally includes a reduction in the number of vehicles and containers involved in the process, and route optimization [89].

An important factor associated with waste collection and transport is urban traffic jams that interfere with the flow of the service fleet. Furthermore, garbage truck circulation can be responsible for delays in traffic flow and consequently generate additional congestion [90], especially when going through many roundabouts to the landfill [84].

Another consideration is economics, where the aim is to minimize the cost of collection and transport [23]. Economopoulou et al. [85] analyzed the contribution of waste transport, keeping in mind the minimization of annualized capital investment and annual operating cost of the entire waste treatment chain and taking into account financial incentives or disincentives, e.g., possible revenue from selling products or services. Concerning waste management systems, Das and Bhattacharyya [91] proposed an optimal waste collection and transport scheme focused on minimizing route length and considering the costs and profits involved.

Analyzing the environmental aspects of waste collection and transport, some studies considered factors such as route selection and optimization, as well as route length [23,92]. Several studies also addressed energy consumption and the emission of air pollutants, particularly CO₂, as an important indicator to be considered because of climate change [93–95]. Bektas and Laporte [93] presented an extension of the classical vehicle routing problem, with a broader objective function that includes total GHG emissions, and Demir et al. [89] performed a comparative analysis of several vehicle emission models to identify ways to minimize the harmful effects of road freight transport on the environment.

15 of 26

Moreover, transport planning from the point of waste collection to the landfill needs to pay particular attention to the correct assessment of the whole burden of the waste management systems [90]. Calabrò et al. [96] demonstrated that the effectiveness of waste management systems strongly depend on an integrated system that presents efficient separate collection, high energy recovery, and extremely limited landfill disposal. Under these circumstances, Chi et al. [97], e.g., analyzed the importance of the separate collection to the entire environmental performance of a waste system, and Pérez et al. [98], using the life cycle assessment (LCA) methodology, presented a calculation of the impact on climate change related to waste collection vehicles.

Another important issue that must be highlighted is the responsibility of the goods supply sector to promote reverse logistics for the customer, thus managing collection and return of any damaged or defective goods due to the need to exchange or the desire to cancel the purchase [99]. The use of technology, such as an omnichannel, which corresponds to a digital connection mechanism with consumers, is an example of a best practice that seeks to facilitate the return of products immediately in the delivery phase or from an agreed date and time, thereby, recovering products unsuitable for an intended purpose and at the same time reducing the number of unnecessary trips for collection [100].

It is also important to note that some best practices applied to distribution and transport can also be applied to collection and transport, such as night-time collection and distribution and the use of information systems to track and monitor the fleet, discussed in Section 5.1.4.

In addition, transport-related problems include waste collection, making it crucial to also consider indirect transport through delivery operations between transfer stations to and from the processing plant [86]. Thus, computational optimization techniques are fundamental to the development of a robust transfer station grid, which can be designed to deal with all possible realizations of future projects of transport infrastructure and technological solutions [101], as well as to optimize waste routing between the existing transfer stations and processing plants [91].

5.1.7. Waste and Recycle and Emissions

In the progress towards more sustainable urban policies, the waste management system is a highly challenging issue, because many regions worldwide use landfilling as a main waste disposal method, especially in developing countries [102]. Thus, waste management plays a fundamental role in a CE as it establishes a long-term path with long-term targets for decreasing the volume of landfilled waste and increasing recycling and reuse [90].

A key factor for proper waste management is knowledge about waste production and its trends. Specifically, regarding the generation of waste by road transport, the continuously growing population and urbanization and the intensive use of private vehicles seen in recent decades is causing waste from the ELVs to increase significantly [103], including items such as end-of-life tires [104]. There is, in addition, the rising consumption of resources used for vehicle propulsion and the resulting environmental pollution [105].

In order to assist with waste management, the United Nations highlights five key factors: (i) establishment of policies and regulations, (ii) support of institutions, (iii) adequate financial mechanisms, (iv) stakeholder participation, and (v) supporting technologies, which can be incorporated through diverse strategies, such as remanufacturing or retreading, and through the integration of the structure of reverse logistics [106].

In waste management, the major current environmental issues to be considered include not only proper treatment and disposal of waste but also the management of GHG generation by the system [107] and related cost impact [108]. Volume and physical composition of waste must be considered when exploring the correlation between urban waste management and GHG emissions due to differences in local environments and lifestyles that lead to a frequent variation in its amount and composition [109]. However, improving the waste management system is an expensive budget item, especially if adopted in developing countries [110].

Thus, waste treatment methods must be studied at the local level, considering facilities and peculiarities present in the region under analysis [109], as well as its budget for investment [110]. Furthermore, almost all waste management stages generate GHG emissions, making it necessary to design appropriate treatment methods, from source to final disposal, to reduce their environmental impact [107]. Specifically, for road transport, the impact on emissions from vehicles of waste collection is related to transport efficiency improvement and energy consumption reduction, which will permit urban solid waste management to achieve sustainability goals [109].

The environmental impact of waste management systems should be properly considered by authorities of both small and large cities to seek improvements, and thus fulfill waste policy targets [102]. The availability of information supported by valid and reliable methods is an essential condition for assessing the impact of these systems. LCA methodology is among those most commonly used for this purpose [90]. In addition, the concept of CE is complementary to key aspects on LCA, considering impacts and in turn solutions across the relevant systems [111].

In the transport sector, LCA research considers all of the environmentally significant processes throughout the vehicle life cycle [112], involving raw material extraction, component manufacturing, assembly, transport, distribution, vehicle use, and end-of-life treatment [113]. Thus, different propulsion technologies, including internal combustion, hybrid, plug-in hybrid, and 100% electric, were presented in various studies that focused on evaluating the eco-profile of vehicles [112].

For example, electric and hybrid engines are promising emerging technologies for propelling vehicles with the potential to reduce GHG emissions from road transport [114] since they are more energy-efficient than conventional internal combustion powertrains. In addition, with fully electric propulsion, there is zero direct emission, and thus no need for a tailpipe [115].

Electric vehicles also have considerably fewer parts and components than those propelled by a combustion engine. Thus, there is no necessity for any type of oil change, nor coolant water, nor sparking device, thus allowing less wear on the vehicle's engine components [116] and, consequently, prolonging their lifetime. These characteristics show that electric vehicles are an appropriate alternative for reducing emissions and waste generation.

Another technological alternative used to reduce GHG emissions from road transport is the use of hydrogen fuel cell vehicles. Hydrogen is an efficient clean energy carrier; thus, the fuel cell can generate direct current power to drive the vehicle. A more sustainable hybrid vehicle can be produced by integrating batteries and control system strategies along with a hydrogen fuel cell [117].

It is noteworthy that an adequate environmental assessment of different propulsion technologies (e.g., hybrid, plug-in hybrid, and 100% electric vehicles) requires investigation of all vehicle LCA stages (production, use, treatment, and disposal) across a wide range of impact categories [112] that represent environmental issues of concern.

Regarding emissions from the combustion of fossil fuels, carbon fixation from the atmosphere is possible through compensating actions, such as reforestation. However, more incisive technologies can be considered. According to WRI [118], direct air capture technologies aren't aimed at reducing emissions from industrial processes as proposed by carbon capture and storage (CCS) but remove carbon dioxide directly from the ambient air by chemical scrubbing.

Biofuels used in transport, such as sugar cane ethanol or biodiesel, have a closed cycle for GHG emissions and are considered the most suitable alternative fuel to displace the use of petroleum-based fuel. However, this closed-cycle does not consider non-carbon-based pollutants. The results of the study developed by Dias et al. [119] and Glensor et al. [120] show that the use of biofuels increased emissions of NO_x . Additionally, Moreira et al. [121] inferred that augmentation of biodiesel blended in with diesel increases emissions of toxic elements that impact human health.

In addition, it should be mentioned that waste prevention through "Reduce, Reuse and Recycle" (3 Rs) represents one of the main directions of waste management in terms of CE, promoting the adoption of closing-the-loop production patterns within an economic system [122].

Since other Rs were addressed in previous sections, specifically regarding recycling, it is important to elaborate public policies and regulations for the development of an ELV collection and recycling network that covers most of the territory, promoting the proper dismantling of vehicles in authorized and well-distributed facilities [103].

Polices oriented to implement recycling practices is also important, such as the ELV Recycling Law [123] in Japan, created to promote waste collection and recovery by vehicle manufacturers and importers during the ELV recycling process, and also to apply recycling fees to be paid by vehicle owners and to provide processors registration and licensing [122], creating a new recycling system for proper processing and disposal of ELVs and their efficient use as resources [124].

Overall, main variations in environmental impact generated by vehicles have their origin in the distance between a disassembly plant and recycling plant and the nature of the vehicle components, such as the mass of the engine, fuel tank, parts, tires, wheels, battery, and remaining wreckage [125]. Specifically, for vehicle components, the recycling rate can be improved following separation and material selection techniques considered [126].

Ferrão and Amaral [127] indicated that separation of 68 parts, equivalent to 14% of vehicle mass and corresponding to a recycling rate of more than 80%, requires a total time of around one hour per vehicle and two employees dedicated to separating the components. In addition, gradual target regulation measures can be adopted for a more efficient recycling process, such as establishing a vehicle recovery rate [126].

Furthermore, according to Soo et al. [49], the steel recovery process through recycling and purification is responsible for increasing the environmental impact of its life cycle by 68% upon GHG-emission-related climate change. This indicates that even the recycling step in waste management can contribute to climate change and that investment should be directed towards new technologies for material recovery.

To solve, or at least minimize, problems from waste production and to stimulate the processes of prevention, recycling, energy, and material recovery in its disposal, it is necessary to promote system-wide planning within the supply chain, including the proper processing of waste and the definition of a structure for waste collection and transport. This could promote efficiency to reduce both total system costs and GHG emissions, based on actions for design optimization of municipal solid waste treatment related to waste management hierarchy, such as the method proposed by Šomplák et al. [128].

In addition, a strategic implementation of measures used to improve waste transport included in the CE stages can be defined based on (i) regulations; (ii) analysis of technical competence requirements for transport operations, waste management, and description; (iii) instructions for waste collection and transport in emergencies as defined by the Secretary of State; and (iv) encouraging the use of innovative technologies for a waste treatment capable of generating fuels and complying with laws such as the RTFO created in the United Kingdom [129].

5.1.8. Disposal and Treatment

Waste disposal has been a critical issue since the industrial revolution due to a significant increase in waste generation in cities [130]. According to IRP [131], material consumed by cities will increase from 40 billion tons in 2010 to 90 billion tons by 2050, which is more than the planet can sustainably accommodate. Thus, it is necessary to put into practice actions to minimize the adverse effects of the waste disposal process and its impacts on the environment [130]. Research and analysis of the actual situation of the management of ELVs is also important. An ELV is defined as a vehicle that is discarded by its registered owner as waste [132]. There are two losses involved in the absence of proper recycling and putting all the garbage in landfills: cost of disposal and cost of not selling this garbage [133]. For ELVs and oil residue disposal management, national guidelines and regulations can be developed and applied as a means of reducing environmental impacts [134].

Moreover, regulation development can be used as a best practice to forbid illegal channels of garbage collection and inadequate disposal, and allow manufacturers aligned with recycling facilities to recycle their products and those of competitors [26]. Citing an example of regulation in the transport sector, European Directive 2000/53/EC [135] has waste reduction that comes from disposal as its purpose and establishes goals for reuse, recycling, and recovery in vehicles' end-of-life phase [136].

About 75% of vehicle components are recyclable, especially metals, including aluminum, while 25% are characterized as hazardous waste [137]. In Europe, the major activity of the vehicle recycling system is metal recycling [135]. In Japan, car recycling has reached 95% [138]; in China, where the focus is mainly on scrap metal recovery and recycling, the rate is less than 70% [122]. In emergent countries, such as Brazil, only 1.5% of the fleet leaving circulation is recycled [139].

It is important to establish ELV management to avoid environmental pollution and to recover useful materials, mainly metals. Fluids and hazardous components, such as batteries, must be removed, then reused, and recycled. As opposed to metals, which can be easily and profitably be sorted, non-metallic residue, known as "car fluff" or "automotive shredder residue" (ASR), ends up in landfills in many European countries [140].

After metals, plastics are the most used vehicle material and comprise 7–9.3% of the vehicle mass [52]. The percentage of plastic is increasing in newer vehicles [141]. Vehicles are being dismantled at shredder facilities that recover some materials, and also produce ASR [137]. ASR contains plastics, rubber, textile and fiber material, wood, and glass, which cannot be recycled and represents 20–25% of vehicle mass [52]. ASR can be classified as heavy fluff or light fluff that comprises about 75% of ASR. Due to its heterogeneity of substances, ASR must be characterized before being recovered [52].

Addressing this need, a shredder trial campaign was developed and performed in Italy in 2008, aimed at verifying the achievable productivity and energy recovery efficiency of ELV reverse supply chain by a careful application of best practices of dismantling. In the final separation of metals and before automatic crushing, it was detected that 8% of metals and 40% of polymers could still be recovered and, in order to optimize this performance, greater standardization is required in the procedure of dismantling and also in the technologies adopted [140].

According to Vermeulen et al. [52], energy can be recovered from the heat created during the incineration of combustible waste. Incineration as a solution to reduce the volume of waste [13] is often the preferred choice in countries lacking space for waste disposal, especially in developed countries such as Switzerland, Japan, France, Germany, Sweden, and Denmark [142]. However, hazardous waste incineration is not viable in these places [143]. ASR is considered a hazardous material in Europe if it is classified as containing hazardous substances [144] and cannot be used in energy recovery, save for when flue gas treatment is used to avoid the emission of hazardous pollutants [52]. However, most of the ASR in Europe is still landfilled [144].

It should be highlighted here that waste-to-energy (WTE), defined as a treatment process for energy recovery from a waste source—such as mixed municipal waste—in the form of heat, electricity, or fuels useable in transport, can create synergies with energy and climate policy without compromising the achievement of higher rates of reuse and recycling [145]. Specifically regarding supply chains, WTE processes are an effective way to produce energy and, at the same time, minimize the use of landfill space. Thus, WTE supply chains address issues of energy demand, waste management, and GHG emissions in line with the CE system [146].

Some measures used to achieve sustainability of ELV recycling were presented by Berzi et al. [136]. The authors suggested targets for increasing the recycle-recovery rate of ELVs and, consequently, decreasing the amount of waste for disposal and treatment, through the following objectives: reduction of waste generation, particularly hazardous waste (including innovation in vehicle design and forbidding the use of heavy metals, such as mercury and cadmium); availability of nearby collection facilities (including sending the vehicle to authorized treatment facilities, ATS); and treatment of ELVs. This includes reliable storage for avoiding environmental pollution, dismantling (reusing and recycling), utilizing standard coding for materials and making available vehicle-dismantling information, and configuring report and information systems [26].

In addition, it is necessary to consider strategic planning for multi-commodity waste network flow problems. Thus, optimization techniques must be used to minimize the total costs of waste treatment, i.e., the processing cost and the transport cost [147], as well as to reduce their impact on the environment [148].

Somplák et al. [147] used computational optimization to find the minimum total treatment cost of a multi-commodity network flow problem in which waste is transported from waste producers (municipalities) for its final treatment in waste processors (incinerators, mechanical-biological treatment plants, landfills). The model provides an ideal total waste flow along the roads, railways, and other channels.

Building on this work, Somplák et al. [148] proposed a mathematical model to analyze a network flow problem that considers the contribution to global warming potential by waste producers. The model allows the identification of means to minimize total costs of waste treatment and reduce GHG emissions. Therefore, the method can contribute to achieving targets for emission reduction in individual regions and for specific producers.

5.2. Transportation Best Practices and Their Association to the Sustainable Development Goals

As discussed in Section 5.1, it is noted that CE can be applied to road transport to promote climate change mitigation and reduce resource depletion through the incorporation of best practices. Throughout all that has been presented, it is important to emphasize the importance of keeping resources extracted and transformed into products in the system in order to close the loop and achieve the integration of all CE stages.

Furthermore, the best practices identified in this paper are related to SDGs and the adoption of a sustainable transport system. In this context, encouraging the use of second life batteries and other more durable and better functioning and modular products in the transport sector is aligned with "responsible consumption and production" (Goal 12). Likewise, route optimization, efficient logistics, tolls to fight traffic congestion, and vehicle occupancy optimization contribute to "sustainable cities and communities" (Goal 11).

In the same context, reducing energy consumption in vehicle manufacturing and the RTFO are aligned with "affordable and clean energy" (Goal 7). Finally, actions such as guidelines and regulations for the disposal of ELVs and oil residues are best practices in line with "climate action" (Goal 13). In addition, it is essential to pay special attention to waste that cannot be treated and returned to the cycle and for which the only possible solution is disposal, providing a minimal impact on the environment, either in the emission of GHG or in its disposal in landfills.

In this study, the role of GHG emissions and air pollutant management is highlighted in the context of CE for transport. In addition, emissions from transport operations, which, unlike changes in vehicle parts and components, involve changes that are too complex to be conducted within a closed loop. Studies regarding pollutant emissions have pointed out the need for an extensive transition to more efficient and sustainable propulsion systems, capture technologies, and biofuel use.

During the literature review, studies on the CE applications in road transport focused mainly on stages that close the loop, such as waste and its disposal and treatment, disregarding other CE stages. Giving equal importance to all CE stages is needed to address all impacts on the road transport sector. Considering high energy consumption and GHG emissions by the transport sector, CE stages oriented toward transport as well as the best practices that face these impacts are of prime importance. The contributions of this study remain in providing consolidated

are of prime importance. The contributions of this study remain in providing consolidated qualitative information from the literature to support transport decision makers in implementing ways to minimize waste and pollution emissions from the transport sector. In this context, to prioritize best practices through quantitative methods, Assis et al. [149] developed a methodology that applies SWOT analysis, sustainability balanced scorecard (SBSC), and the analytic hierarchy process (AHP), which could be applied to select best practices of each CE stage.

6. Conclusions

This paper addresses the CE concept for the transport sector and presents best practices that can support strategies to assist decision makers, given that strategy is still lacking in the literature. The concepts are divided into the first and second life cycles, presenting opportunities to close the loop. This study focuses on the urban scale since cities are major emitters of GHG and pollutants due to rapid economic growth and urbanization.

Important knowledge gaps were identified due to the limitations of existing studies investigating the subject. The main gap identified in this paper was the lack of studies that address all cycle stages. In general, there are studies focused mainly on the final CE stages, such as waste disposal and treatment. In addition, few studies considered transport as the main topic or presented the 7 Rs in the reverse cycle. Transport was only being used as a tool to optimize some processes, such as distribution and collection.

The proposed application of CE stages to the urban transport sector and the related best practices (Section 5) can assist transport decision makers and local and regional authorities in policymaking to develop strategies that reduce resource depletion, and in promoting measures for climate change mitigation. In addition, the closed loop of transport system must be encouraged.

Mobility solutions must take into account the economic, environmental, and societal aspects of sustainability as a model that contributes to improving people's quality of life, which is directly related to the needs and limits of natural resources. The current linear economy model is unsustainable; thus, moving toward a CE is becoming increasingly important. Value can be created due to the rational use of resources and minimization of environmental impact at all phases of the product life cycle in the transport sector.

CE best practices applied to the transport sector, obtained as a result of this study, could be considered by countries' efforts to reduce GHG emissions toward their Nationally Determined Contributions (NDCs) as a central element for implementing the Paris Agreement. In addition, it contributes reaching Sustainable Development Goals (SDGs), such as "affordable and clean energy" (Goal 7), "sustainable cities and communities" (Goal 11), "responsible consumption and production" (Goal 12), and "climate action" (Goal 13).

It is also worth mentioning, as a limitation of this study, that quantitative analyses were not considered, i.e., ones that do not compare and/or evaluate reduced amounts of resources or mitigated emissions from performance indicators, but a qualitative analysis is performed based on best practices applied to each CE stage identified throughout the literature review. A method to prioritize best practices should be applied to quantitative analysis, identified specifically for the CE stage wherein it will be adopted.

As a suggestion for future research, a further review can be done to establish indicators for quantifying the benefits of implementing the best practices at each of the CE stages, as well as identifying potential constraints for their application or implementation to fill the knowledge gaps identified. Further studies could be elaborated to encompass other transportation modes and their integration system. Another key aspect is to ascertain how new technologies implemented in the transport sector, such as how electric or hydrogen-powered vehicles can promote benefits for each stage of the CE, i.e., to use these advances not just as a best practice, but as the main objective under analysis.

Author Contributions: Conceptualization, V.H.S.D.A., M.G.D.C., V.X.D.C. and T.F.D.A.; investigation, V.H.S.D.A., M.G.D.C., V.X.D.C. and T.F.D.A.; methodology, V.H.S.D.A., M.G.D.C., V.X.D.C. and T.F.D.A.; resources, A.S.S. and M.d.A.D.; supervision, A.S.S. and M.d.A.D.; writing—original draft, V.H.S.D.A., M.G.D.C., V.X.D.C. and T.F.D.A.; writing—review and editing, V.H.S.D.A., M.G.D.C., V.X.D.C. and T.F.D.A. All authors have read and agreed to the published version of the manuscript.

Funding: This study was financed in part by the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior, Brasil (CAPES), Finance Code 001. This work was partially supported by the National Council for Scientific and Technological Development (CNPq), under grant 423127/2018-7. This work was supported by Carlos Chagas Filho Foundation for Research Support of the State of Rio de Janeiro, under grant #2021007191.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Ngan, S.L.; How, B.S.; Teng, S.Y.; Promentilla, M.A.B.; Yatim, P.; Er, A.C.; Lam, H.L. Prioritization of sustainability indicators for promoting the circular economy: The case of developing countries. *Renew. Sustain. Energy Rev.* **2019**, *111*, 314–331. [CrossRef]
- Van Fan, Y.; Lee, C.T.; Lim, J.S.; Klemeš, J.J.; Le, P.T.K. Cross-disciplinary approaches towards smart, resilient and sustainable circular economy. J. Clean. Prod. 2019, 232, 1482–1491. [CrossRef]
- United Nations. Paris Agreement; United Nations: New York, NY, USA, 2015; Available online: https://unfccc.int/files/essential_ background/convention/application/pdf/english_paris_agreement.pdf (accessed on 10 June 2020).
- 4. IPCC Global Warming of 1.5 °C. Available online: https://www.ipcc.ch/sr15/ (accessed on 24 July 2020).
- Mountford, H.; Waskow, D.; Gonzalez, L.; Gajjar, C.; Cogswell, N.; Holt, M.; Fransen, T.; Bergen, M.; Gerholdt, R. COP26: Key Outcomes from the UN Climate Talks in Glasgow. Available online: https://www.wri.org/insights/cop26-key-outcomes-unclimate-talks-glasgow (accessed on 15 June 2022).
- 6. Arora, N.K.; Mishra, I. COP26: More challenges than achievements. *Environ. Sustain.* 2021, 4, 585–588. [CrossRef]
- Santos, A.S.; de Abreu, V.H.S.; de Assis, T.F.; Ribeiro, S.K.; Ribeiro, G.M. An overview on costs of shifting to sustainable road transport: A challenge for cities worldwide. In *Carbon Footprint Case Studies*; Environmental Footprints and Eco-Design of Products and Processes Book Series; Springer: Singapore, 2021; pp. 93–121. [CrossRef]
- 8. Ellen MacArthur Foundation Circular Economy. *Cities-Urban Mobility System*; Ellen MacArthur Foundation Circular Economy: Cowes, UK, 2012; Volume 66.
- 9. IEA. Energy Efficiency Indicators: Highlights; International Energy Agency: Paris, France, 2016.
- United Nations. 68% of the World Population Projected to Live in Urban Areas by 2050, Says UN; United Nations: New York, NY, USA, 2018; Available online: https://www.un.org/development/desa/en/news/population/2018-revision-of-world-urbanization-prospects.html (accessed on 24 July 2020).
- 11. IEA. Tracking Transport 2020. Available online: https://www.iea.org/reports/tracking-transport-2020 (accessed on 23 June 2020).
- Climate Chance Association. Climate Chance Association. Climate change. In Sector-Based Book-Synthesis Report on Climate Change, 1st ed.; Climate Chance Association: Paris, France, 2019; Available online: https://www.climate-chance.org/wp-content/ uploads/2020/03/climate-chance-2019-sector-based-book-2019-synthesis-report-on-climate-action-by-sector.pdf (accessed on 5 April 2020).
- 13. Seroka-Stolka, O.; Ociepa-Kubicka, A. Green logistics and circular economy. Transp. Res. Procedia 2019, 39, 471–479. [CrossRef]
- 14. Santos, A.S. Circular Economy–A Pathway to Achieve Sustainable Development. Int. J. Environ. 2019, 8. [CrossRef]
- 15. Walmsley, T.G.; Ong, B.H.Y.; Klemeš, J.J.; Tan, R.R.; Varbanov, P.S. Circular Integration of processes, industries, and economies. *Renew. Sustain. Energy Rev.* **2019**, *107*, 507–515. [CrossRef]
- 16. Mitchell, P.; James, K. Economic growth potential of more circular economies. In Proceedings of the ISWA World Congress, Antwerp, Belgium, 7–9 September 2015; pp. 1–28. [CrossRef]
- 17. IEA. *Global EV Outlook*; International Energy Agency: Paris, France, 2021; Available online: https://www.iea.org/reports/globalev-outlook-2021 (accessed on 20 June 2021).
- 18. IEA. Transport Biofuels-COVID-19 Causes the First Contraction in Biofuel Output in two Decades. Available online: https://www.iea.org/reports/renewables-2020/transport-biofuels (accessed on 20 July 2021).
- 19. Haddach, A.; Ammari, M.; Allal, L.B. Measuring Sustainability of Best Practices in Logistics Chains. *Glob. J. Sci. Front. Res.* 2016, 16, 12–23.
- United Nations. Transforming Our World: The 2030 Agenda for Sustainable Development-A/RES/70/1; United Nations: New York, NY, USA, 2015; Available online: https://www.un.org/en/development/desa/population/migration/generalassembly/docs/ globalcompact/A_RES_70_1_E.pdf (accessed on 28 April 2020).

- Kirchherr, J.; Reike, D.; Hekkert, M. Conceptualizing the circular economy: An analysis of 114 definitions. *Resour. Conserv. Recycl.* 2017, 127, 221–232. [CrossRef]
- 22. Millar, N.; McLaughlin, E.; Börger, T. The Circular Economy: Swings and Roundabouts? Ecol. Econ. 2019, 158, 11–19. [CrossRef]
- 23. Nowakowski, P.; Mrówczyńska, B. Towards sustainable WEEE collection and transportation methods in circular economy-Comparative study for rural and urban settlements. *Resour. Conserv. Recycl.* **2018**, 135, 93–107. [CrossRef]
- Sariatli, F. Linear Economy Versus Circular Economy: A Comparative and Analyzer Study for Optimization of Economy for Sustainability. Visegr. J. Bioecon. Sustain. Dev. 2017, 6, 31–34. [CrossRef]
- Wautelet, T.; Impakt, P. The Concept of Circular Economy: Its Origins and its Evolution. 2018, pp. 1–30. Available online: https://doi.org/10.13140/RG.2.2.17021.87523 (accessed on 6 May 2022).
- Saidani, M.; Kendall, A.; Yannou, B.; Leroy, Y.; Cluzel, F. Management of the end-of-life of light and heavy vehicles in the U.S.: Comparison with the European union in a circular economy perspective. *J. Mater. Cycles Waste Manag.* 2019, 21, 1449–1461. [CrossRef]
- 27. Ghisellini, P.; Cialani, C.; Ulgiati, S. A review on circular economy: The expected transition to a balanced interplay of environmental and economic systems. *J. Clean. Prod.* 2015, *114*, 11–32. [CrossRef]
- Murray, A.; Skene, K.; Haynes, K. The Circular Economy: An Interdisciplinary Exploration of the Concept and Application in a Global Context. J. Bus. Ethics 2015, 140, 369–380. [CrossRef]
- Suárez-Eiroa, B.; Fernández, E.; Méndez-Martínez, G.; Soto-Oñate, D. Operational principles of circular economy for sustainable development: Linking theory and practice. J. Clean. Prod. 2019, 214, 952–961. [CrossRef]
- 30. Saidani, M.; Kendall, A.; Yannou, B.; Leroy, Y.; Cluzel, F. What about the circular economy of vehicles in the U.S.? An extension of the analysis done in the EU by Saidani et al. (2017). *Resour. Conserv. Recycl.* **2018**, *136*, 287–288. [CrossRef]
- OECD. Material Resources, Productivity and the Environment: Key Findings; Organization for Economic Co-Operation and Development: Paris, France, 2013; Available online: https://www.oecd.org/greengrowth/MATERIAL%20RESOURCES,%2 0PRODUCTIVITY%20AND%20THE%20ENVIRONMENT_key%20findings.pdf (accessed on 28 April 2020).
- 32. IEA. Material Efficiency in Clean Energy Transitions; International Energy Agency: Paris, France, 2019. [CrossRef]
- 33. European Commission. *Ecodesign Working Plan 2016–2019*; European Commission: Brussels, Belgium, 2016; Available online: https://ec.europa.eu/energy/sites/ener/files/documents/com_2016_773.en_.pdf (accessed on 10 June 2020).
- 34. Clark, G. Evolution of the global sustainable consumption and production policy and the United Nations Environment Programme's (UNEP) supporting activities. J. Clean. Prod. 2007, 15, 492–498. [CrossRef]
- 35. Fowler, K.R. Logistics, Distribution, and Support. In *Developing and Managing Embedded Systems and Products*; Elsevier Inc.: Amsterdam, The Netherlands, 2015; pp. 649–672. ISBN 9780124058798.
- 36. Camacho-Otero, J.; Boks, C.; Pettersen, I.N. Consumption in the circular economy: A literature review. *Sustainability* **2018**, *10*, 2758. [CrossRef]
- EPA. Sustainable Materials Management: Non-Hazardous Materials and Waste Management Hierarchy. Available online: https://www.epa.gov/smm/sustainable-materials-management-non-hazardous-materials-and-waste-managementhierarchy#Treatment (accessed on 31 July 2021).
- Ellen MacArthur Foundation. Towards the Circular Economy-Economic and Business Rationale for an Accelerated Transition. 2013. Available online: https://emf.thirdlight.com/link/ip2fh05h21it-6nvypm/@/preview/1?o (accessed on 28 April 2020).
- 39. Goes, G.V.; Gonçalves, D.N.S.; D'Agosto, M.d.A.; La Rovere, E.L.; Albergaria, R.B.D.M. MRV framework and prospective scenarios to monitor and ratchet up Brazilian transport mitigation targets. *Clim. Chang.* **2020**, *162*, 2197–2217. [CrossRef]
- EU. Directive 2008/122/EC of the European Parliament and of the Council on waste and repealing certain Directives. *Fundam. Texts Eur. Priv. Law* 2008, 3–30. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex%3A32008L0098 (accessed on 18 July 2022).
- Durand, A.; Zijlstra, T.; van Oort, N.; Hoogendoorn-Lanser, S.; Hoogendoorn, S. Access denied? Digital inequality in transport services. *Transp. Rev.* 2021, 42, 1–26. [CrossRef]
- 42. Dolganova, I.; Rödl, A.; Bach, V.; Kaltschmitt, M.; Finkbeiner, M. A review of life cycle assessment studies of electric vehicles with a focus on resource use. *Resources* 2020, *9*, 32. [CrossRef]
- Geissdoerfer, M.; Savaget, P.; Bocken, N.M.P.; Hultink, E.J. The Circular Economy–A new sustainability paradigm? J. Clean. Prod. 2016, 143, 757–768. [CrossRef]
- 44. Bauer, C.; Hofer, J.; Althaus, H.J.; Del Duce, A.; Simons, A. The environmental performance of current and future passenger vehicles: Life Cycle Assessment based on a novel scenario analysis framework. *Appl. Energy* **2015**, 157, 871–883. [CrossRef]
- International Organization for Standardization. ISO 14044-Environmental Management—Life Cycle Assessment; International Organization for Standardization: Geneva, Switzerland, 2006; Available online: https://www.iso.org/standard/38498.htm (accessed on 10 June 2020).
- 46. Al-Sheyadi, A.; Muyldermans, L.; Kauppi, K. The complementarity of green supply chain management practices and the impact on environmental performance. *J. Environ. Manag.* 2019, 242, 186–198. [CrossRef]
- 47. Hu, J.; Liu, Y.L.; Yuen, T.W.W.; Lim, M.K.; Hu, J. Do green practices really attract customers? The sharing economy from the sustainable supply chain management perspective. *Resour. Conserv. Recycl.* **2019**, 149, 177–187. [CrossRef]
- El-Haggar, S.M. Cleaner production. In Sustainable Industrial Design and Waste Management Attitudes; Elsevier: Amsterdam, The Netherlands, 2007; ISBN 978-0-12-373623-9.

- 49. Soo, V.K.; Compston, P.; Doolan, M. Is the Australian Automotive Recycling Industry Heading towards a Global Circular Economy?-A Case Study on Vehicle Doors. *Procedia CIRP* **2016**, *48*, 10–15. [CrossRef]
- 50. Rogers, R. Cities for a Small Planet; GGili: Barcelona, Spain, 2001; ISBN 84-252-1889-6.
- 51. Koroneos, C.J.; Achillas, C.; Moussiopoulos, N.; Nanaki, E.A. Life Cycle Thinking in the Use of Natural Resources. *Open Environ. Sci.* **2013**, *7*, 1–6. [CrossRef]
- Vermeulen, I.; Van Caneghem, J.; Block, C.; Baeyens, J.; Vandecasteele, C. Automotive shredder residue (ASR): Reviewing its production from end-of-life vehicles (ELVs) and its recycling, energy or chemicals ' valorisation. *J. Hazard. Mater.* 2011, 190, 8–27. [CrossRef]
- 53. Li, L.; Loo, B.P.Y. Alternative and Transitional Energy Sources for Urban Transportation. *Curr. Sustain. Energy Rep.* **2014**, *1*, 19–26. [CrossRef]
- Pan, Y.; Li, H. Sustainability evaluation of end-of-life vehicle recycling based on emergy analysis: A case study of an end-of-life vehicle recycling enterprise in China. In Proceedings of the 9th Biennial Emergy Conference, Gainesville, FL, USA, 6–9 January 2017; pp. 243–256. Available online: https://cep.ees.ufl.edu/emergy/documents/conferences/ERC09_2016/32_Pan_Li.pdf (accessed on 28 April 2020).
- Aguilar Esteva, L.C.; Kasliwal, A.; Kinzler, M.S.; Kim, H.C.; Keoleian, G.A. Circular economy framework for automobiles: Closing energy and material loops. J. Ind. Ecol. 2021, 25, 877–889. [CrossRef]
- 56. Cusenza, M.A.; Guarino, F.; Longo, S.; Ferraro, M.; Cellura, M. Energy and environmental benefits of circular economy strategies: The case study of reusing used batteries from electric vehicles. *J. Energy Storage* **2019**, *25*, 100845. [CrossRef]
- 57. Faussone, G.C. Transportation fuel from plastic: Two cases of study. Waste Manag. 2017, 73, 416–423. [CrossRef]
- 58. Wang, H.; Wang, T.; Johnson, L.A.; Pometto III, A.L. Effect of the Corn Breaking Method on Oil Distribution between Stillage Phases of Dry-Grind Corn Ethanol Production. *J. Agric. Food Chem.* **2008**, *56*, 9975–9980. [CrossRef]
- 59. Kerr, B.J.; Dozier, W.A.; Shurson, G.C. Lipid digestibility and energy content of distillers' corn oil in swine and poultry 1. *J. Anim. Sci.* **2016**, *94*, 2900–2908. [CrossRef] [PubMed]
- IEA. Tracking Industry. 2020. Available online: https://www.iea.org/reports/tracking-industry-2020 (accessed on 22 September 2020).
- 61. Stephan, B.; Lee, I.; Kim, J. Crashing the Climate: How the Car Industry Is Driving the Climate Crisis. 2019. Available online: https://issuu.com/greenpeace_eastasia/docs/crashing_the_climate_how_the_car_industry_is_drivi (accessed on 11 June 2020).
- 62. Nakamoto, Y.; Nishijima, D.; Kagawa, S. The role of vehicle lifetime extensions of countries on global CO₂ emissions. *J. Clean. Prod.* **2019**, 207, 1040–1046. [CrossRef]
- 63. Deif, A.M. A system model for green manufacturing. J. Clean. Prod. 2011, 19, 1553–1559. [CrossRef]
- 64. Kieckhäfer, K.; Wachter, K.; Spengler, T.S. Analyzing manufacturers' impact on green products' market diffusion-the case of electric vehicles. *J. Clean. Prod.* 2016, 162, 1–15. [CrossRef]
- 65. Govindan, K.; Soleimani, H. A review of reverse logistics and closed-loop supply chains: A Journal of Cleaner Production focus. J. Clean. Prod. 2017, 142, 371–384. [CrossRef]
- Salem, A.H.; Deif, A.M. Developing a Greenometer for Green Manufacturing Assessment. J. Clean. Prod. 2017, 154, 413–423. [CrossRef]
- 67. Silva, D.A.L.; de Oliveira, J.A.; de Oliveira, R.A.P.F.J.F.G.; da Silva, E.J.; Ometto, A.R. Life Cycle Assessment in automotive sector: A case study for engine valves towards cleaner production. *J. Clean. Prod.* **2018**, *184*, 286–300. [CrossRef]
- Gerrard, J.; Kandlikar, M. Is European end-of-life vehicle legislation living up to expectations? Assessing the impact of the ELV Directive on 'green' innovation and vehicle recovery. J. Clean. Prod. 2007, 15, 17–27. [CrossRef]
- 69. Xiang, W.; Ming, C. Implementing extended producer responsibility: Vehicle remanufacturing in China. *J. Clean. Prod.* 2011, 19, 680–686. [CrossRef]
- 70. Saidani, M.; Yannou, B.; Leroy, Y.; Cluzel, F. Heavy vehicles on the road towards the circular economy: Analysis and comparison with the automotive industry. *Resour. Conserv. Recycl.* 2017, 135, 108–122. [CrossRef]
- 71. Bekrar, A.; El Cadi, A.A.; Todosijevic, R.; Sarkis, J. Digitalizing the closing-of-the-loop for supply chains: A transportation and blockchain perspective. *Sustainability* **2021**, *13*, 2829. [CrossRef]
- Mestre, A.; Cooper, T. Circular Product Design. A Multiple Loops Life Cycle Design Approach for the Circular Economy. *Des. J.* 2017, 20, S1620–S1635. [CrossRef]
- 73. Carrasco-Gallego, R.; Ponce-Cueto, E.; Dekker, R. Closed-loop supply chains of reusable articles: A typology grounded on case studies. *Int. J. Prod. Res.* 2012, *50*, 5582–5596. [CrossRef]
- 74. Barbosa, I.D.C.; Borbon-Galvez, Y.; Verlinden, T.; Van de Voorde, E.; Vanelslander, T.; Dewulf, W. City logistics, urban goods distribution and last mile delivery and collection. *Compet. Regul. Netw. Ind.* **2017**, *18*, 22–43. [CrossRef]
- Gogola, M. Environmental Management & Audit: Tempus Project Recoaud. 4, Environmental Assessment-Featured Articles; SPH— Scientific Publishing Hub: Celje, Slovenia, 2016; ISBN 978-961-6948-15-9.
- 76. Ellen MacArthur Foundation. The Circular Economy in Detail. Available online: https://archive.ellenmacarthurfoundation.org/ explore/the-circular-economy-in-detail (accessed on 22 July 2021).
- Dat, L.Q.; Truc Linh, D.T.; Chou, S.Y.; Yu, V.F. Optimizing reverse logistic costs for recycling end-of-life electrical and electronic products. *Expert Syst. Appl.* 2012, 39, 6380–6387. [CrossRef]

- 78. Lee, S.E.; Quinn, A.D.; Rogers, C.D.F. Advancing city sustainability via its systems of flows: The urban metabolism of birmingham and its hinterland. *Sustainability* **2016**, *8*, 220. [CrossRef]
- Dablanc, L.; Giuliano, G.; Holliday, K.; O'Brien, T. Best practices in urban freight management. *Transp. Res. Rec.* 2013, 2379, 29–38.
 [CrossRef]
- 80. Lah, O. The barriers to low-carbon land-transport and policies to overcome them. Eur. Transp. Res. Rev. 2015, 7, 5. [CrossRef]
- 81. Kolbe, K. Mitigating urban heat island effect and carbon dioxide emissions through different mobility concepts: Comparison of conventional vehicles with electric vehicles, hydrogen vehicles and public transportation. *Transp. Policy* **2019**, *80*, 1–11. [CrossRef]
- 82. Frenken, K.; Schor, J. Putting the sharing economy into perspective. Environ. Innov. Soc. Transit. 2017, 23, 3–10. [CrossRef]
- 83. Ertz, M.; Durif, F.; Arcand, M. Collaborative consumption: Conceptual snapshot at a buzzword. *J. Entrep. Educ.* **2016**, *19*, 1–23. [CrossRef]
- 84. Möhlmann, M. Perceived trustworthiness of online shops. J. Consum. Behav. 2008, 50, 35–50. [CrossRef]
- 85. Skjelvik, J.M.; Erlandsen, A.M.; Haavardsholm, O. *Environmental Impacts and Potential of the Sharing Economy*; Nordic Council of Ministers: Copenhagen, Denmark, 2017; ISBN 9789289351560.
- UNECE. Study on the Application of Energy Efficiency and Renewable Energy Advanced Technologies in Central Asian Countries; UNECE: Kraainem, Belgium, 2013; Available online: https://unece.org/fileadmin/DAM/energy/se/pdfs/gee21/projects/Database_e. pdf (accessed on 10 June 2020).
- 87. ADB. *Reducing Carbon Emissions from Transport Projects*; Asian Development Bank: Manila, Philippines, 2010; Available online: https://www.oecd.org/derec/adb/47170274.pdf (accessed on 28 April 2020).
- Modoi, O.C.; Mihai, F.C. E-Waste and End-of-Life Vehicles Management and Circular Economy Initiatives in Romania. *Energies* 2022, 15, 1120. [CrossRef]
- 89. Kim, B.I.; Kim, S.; Sahoo, S. Waste collection vehicle routing problem with time windows. *Comput. Oper. Res.* 2006, 33, 3624–3642. [CrossRef]
- 90. Peri, G.; Ferrante, P.; La Gennusa, M.; Pianello, C.; Rizzo, G. Greening MSW management systems by saving footprint: The contribution of the waste transportation. *J. Environ. Manag.* **2018**, *219*, 74–83. [CrossRef]
- 91. Das, S.; Bhattacharyya, B.K. Optimization of municipal solid waste collection and transportation routes. *Waste Manag.* **2015**, *43*, 9–18. [CrossRef]
- 92. Tavares, G.; Zsigraiova, Z.; Semiao, V.; Carvalho, M.D.G. A case study of fuel savings through optimisation of MSW transportation routes. *Manag. Environ. Qual. An Int. J.* 2008, 19, 444–454. [CrossRef]
- 93. Bektas, T.; Laporte, G. The Pollution-Routing Problem Tolga Bektas. Transp. Res. Part B 2011, 45, 1232–1250. [CrossRef]
- 94. Demir, E.; Bektas, T.; Laporte, G. A comparative analysis of several vehicle emission models for road freight transportation. *Transp. Res. Part D Transp. Environ.* **2011**, *16*, 347–357. [CrossRef]
- 95. Ahn, K.; Rakha, H. The effects of route choice decisions on vehicle energy consumption and emissions. *Transp. Res. Part D Transp. Environ.* **2008**, *13*, 151–167. [CrossRef]
- 96. Calabrò, P.S.; Gori, M.; Lubello, C. European trends in greenhouse gases emissions from integrated solid waste management. *Environ. Technol.* **2015**, *36*, 2125–2137. [CrossRef]
- Chi, Y.; Dong, J.; Tang, Y.; Huang, Q.; Ni, M. Life cycle assessment of municipal solid waste source-separated collection and integrated waste management systems in Hangzhou, China. J. Mater. Cycles Waste Manag. 2015, 17, 695–706. [CrossRef]
- Pérez, J.; Lumbreras, J.; Rodríguez, E.; Vedrenne, M. A methodology for estimating the carbon footprint of waste collection vehicles under different scenarios: Application to Madrid. *Transp. Res. Part D Transp. Environ.* 2017, 52, 156–171. [CrossRef]
- 99. Panigrahi, S.K.; Kar, F.W.; Fen, T.A.; Hoe, L.K.; Wong, M. A Strategic Initiative for Successful Reverse Logistics Management in Retail Industry. *Glob. Bus. Rev.* 2018, *19*, S151–S175. [CrossRef]
- Kraemer, D. Omni-Channel Logistics: A DHL Perspective on Implications and Use Cases for the Logistics Industry; DHL Customer Solutions & Innovation: Troisdorf, Germany, 2015; Available online: https://ftp.idu.ac.id/wp-content/uploads/ebook/ip/ BUKU%20LOGISTIK%204.0/DHL/dhl_trendreport_omnichannel.pdf (accessed on 14 March 2020).
- Kůdela, J.; Šomplák, R.; Nevrlý, V.; Lipovský, T.; Smejkalová, V.; Dobrovský, L. Multi-objective strategic waste transfer station planning. J. Clean. Prod. 2019, 230, 1294–1304. [CrossRef]
- Abdel-shafy, H.I.; Mansour, M.S.M. Solid waste issue: Sources, composition, disposal, recycling, and valorization. *Egypt. J. Pet.* 2018, 27, 1275–1290. [CrossRef]
- Li, W.; Li, H.; Zhang, H.; Sun, S. The Analysis of CO₂ Emissions and Reduction Potential in China's Transport Sector. *Math. Probl.* Eng. 2016, 2016, 1043717. [CrossRef]
- Gigli, S.; Landi, D.; Germani, M. Cost-benefit analysis of a circular economy project: A study on a recycling system for end-of-life tyres. J. Clean. Prod. 2019, 229, 680–694. [CrossRef]
- 105. Sharma, S.; Basu, S.; Shetti, N.P.; Aminabhavi, T.M. Waste-to-energy nexus for circular economy and environmental protection: Recent trends in hydrogen energy. *Sci. Total Environ.* 2020, 713, 136633. [CrossRef] [PubMed]
- 106. Uriarte-Miranda, M.L.; Caballero-Morales, S.O.; Martinez-Flores, J.L.; Cano-Olivos, P.; Akulova, A.A. Reverse logistic strategy for the management of tire waste in Mexico and Russia: Review and conceptual model. *Sustainability* 2018, 10, 3398. [CrossRef]
- 107. Kristanto, G.A.; Koven, W. Estimating greenhouse gas emissions from municipal solid waste management in Depok, Indonesia. *City Environ. Interact.* **2020**, *4*, 100027. [CrossRef]

- Mavrotas, G.; Gakis, N.; Skoulaxinou, S.; Katsouros, V.; Georgopoulou, E. Municipal solid waste management and energy production: Consideration of external cost through multi-objective optimization and its effect on waste-to-energy solutions. *Renew. Sustain. Energy Rev.* 2015, *51*, 1205–1222. [CrossRef]
- 109. Chen, T.; Lin, C. Greenhouse gases emissions from waste management practices using Life Cycle Inventory model. *J. Harzardous Mater.* **2008**, *155*, 23–31. [CrossRef]
- 110. Zarea, M.A.; Moazed, H.; Ahmadmoazzam, M.; Malekghasemi, S.; Jaafarzadeh, N. Life cycle assessment for municipal solid waste management: A case study from Ahvaz, Iran. *Environ. Monit. Assess.* **2019**, *191*, 131. [CrossRef]
- 111. EEA. Electric Vehicles from Life Cycle and Circular Economy Perspectives-TERM 2018; European Economic Area: Copenhagen, Denmark, 2018.
- 112. Del Pero, F.; Delogu, M.; Pierini, M. Life Cycle Assessment in the automotive sector: A comparative case study of Internal Combustion Engine (ICE) and electric car. *Procedia Struct. Integr.* **2018**, *12*, 521–537. [CrossRef]
- 113. Nordelöf, A.; Messagie, M. Environmental impacts of hybrid, plug-in hybrid, and battery electric vehicles—What can we learn from life cycle assessment ? *Int. J. Life Cycle Assess.* **2014**, *19*, 1866–1890. [CrossRef]
- Nemry, F.; Brons, M. Plug-in Hybrid and Battery Electric Vehicles Market penetration scenarios of electric drive vehicles. *Jrc-Ipts* 2010, 1–36. Available online: https://aetransport.org/public/downloads/j-KZq/4922-514ec606e1709.pdf (accessed on 28 April 2020).
- 115. Sadek, N. Urban electric vehicles: A contemporary business case. Eur. Transp. Res. Rev. 2012, 4, 27–37. [CrossRef]
- 116. Palinski, M. A Comparison of Electric Vehicles and Conventional Automobiles: Costs and Quality Perspective. *Novia* 2017. Available online: https://www.theseus.fi/handle/10024/133032 (accessed on 28 April 2020).
- 117. Manoharan, Y.; Hosseini, S.E.; Butler, B.; Alzhahrani, H.; Senior, B.T.F.; Ashuri, T.; Krohn, J. Hydrogen fuel cell vehicles; Current status and future prospect. *Appl. Sci.* 2019, *9*, 2296. [CrossRef]
- 118. WRI. 6 Ways to Remove Carbon Pollution from the Sky. Available online: https://www.wri.org/blog/2018/09/6-ways-remove-carbon-pollution-sky (accessed on 19 December 2019).
- Dias, G.S.; Fernando de Lima Luz, L., Jr.; Mitchell, D.A.; Krieger, N. Scale-up of biodiesel synthesis in a closed-loop packed-bed bioreactor system using the fermented solid produced by Burkholderia lata LTEB11. Chem. Eng. J. 2017, 316, 341–349. [CrossRef]
- 120. Glensor, K.; Muñoz, B.M.R. Life-cycle assessment of Brazilian transport biofuel and electrification pathways. *Sustainability* **2019**, *11*, 6332. [CrossRef]
- 121. Moreira, C.A.B.; Squizzato, R.; Beal, A.; de Almeida, D.S.; Rudke, A.P.; Ribeiro, M.; Andrade, M.d.F.; Kumar, P.; Martins, L.D. Natural variability in exposure to fine particles and their trace elements during typical workdays in an urban area. *Transp. Res. Part D Transp. Environ.* **2018**, *63*, 333–346. [CrossRef]
- Liu, L.; Liang, Y.; Song, Q.; Li, J. A review of waste prevention through 3R under the concept of circular economy in China. J. Mater. Cycles Waste Manag. 2017, 19, 1314–1323. [CrossRef]
- 123. Ministry of the Environment Enactment of Ministerial Ordinances for Partially Amending the Enforcement Regulations of the Law for the Recycling of End-of- Life Vehicles. Available online: http://www.env.go.jp/en/press/2003/0808b.html (accessed on 2 July 2020).
- Zhao, Q.; Chen, M. A comparison of ELV recycling system in China and Japan and China's strategies. *Resour. Conserv. Recycl.* 2011, 57, 15–21. [CrossRef]
- Li, W.; Bai, H.; Yin, J.; Xu, H. Life cycle assessment of end-of-life vehicle recycling processes in China-Take Corolla taxis for example. J. Clean. Prod. 2016, 117, 176–187. [CrossRef]
- 126. Tian, G.; Chu, J.; Hu, H.; Li, H. Technology innovation system and its integrated structure for automotive components remanufacturing industry development in China. J. Clean. Prod. 2014, 85, 419–432. [CrossRef]
- 127. Ferrão, P.; Amaral, J. Assessing the economics of auto recycling activities in relation to European Union Directive on end of life vehicles. *Technol. Forecast. Soc. Chang.* 2006, *73*, 277–289. [CrossRef]
- Šomplák, R.; Kůdela, J.; Smejkalová, V.; Nevrlý, V.; Pavlas, M.; Hrabec, D. Pricing and advertising strategies in conceptual waste management planning. J. Clean. Prod. 2019, 239, 118068. [CrossRef]
- DEFRA. Our Waste, Our Resources: A Strategy for England; Department for Environment, Food and Rural Affairs: London, UK, 2018. Available online: https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/765914/resources-waste-strategy-dec-2018.pdf (accessed on 28 July 2020).
- 130. Ghalehkhondabi, I.; Maihami, R.; Ahmadi, E. Optimal pricing and environmental improvement for a hazardous waste disposal supply chain with emission penalties. *Util. Policy* **2020**, *62*, 101001. [CrossRef]
- 131. Swilling, M.; Hajer, M.; Baynes, T.; Bergesen, J.; Labbé, F.; Kaviti, J.; Musango, A.; Ramaswami, B.; Robinson, S.S.; Suh, S. *The Weight of Cities: Resource Requirements of Future Urbanization*; United Nations Environment Programme International Resource Panel: Nairobi, Kenya, 2018; Available online: https://wedocs.unep.org/handle/20.500.11822/31624 (accessed on 28 March 2020).
- 132. Simic, V. End-of-life vehicle recycling-a review of the state-of-the-art. Teh. Vjesn. 2013, 20, 371–380.
- 133. Da Silva, C.L. Proposal of a dynamic model to evaluate public policies for the circular economy: Scenarios applied to the municipality of Curitiba. *Waste Manag.* **2018**, *78*, 456–466. [CrossRef]
- 134. Dri, M.; Canfora, P.; Gaudillat, P. Best Environmental Management Practice for the Waste Management Sector; European Union: Luxembourg, 2018. [CrossRef]

- 135. European Parliament. European Parliament and Council of the European Union Directive 2000/53/EC-the "ELV Directive"; European Parliament: Strasbourg, France, 2019; Available online: https://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=COM:2007: 0005:FIN:EN:PDF (accessed on 17 April 2020).
- Berzi, L.; Delogu, M.; Giorgetti, A.; Pierini, M. On-field investigation and process modelling of End-of-Life Vehicles treatment in the context of Italian craft-type Authorized Treatment Facilities. *Waste Manag.* 2013, 33, 892–906. [CrossRef]
- 137. Karagoz, S.; Aydin, N.; Simic, V. End-of-life vehicle management: A comprehensive review. J. Mater. Cycles Waste Manag. 2020, 22, 416–442. [CrossRef]
- 138. Che, J.; Yu, J.S.; Kevin, R.S. End-of-life vehicle recycling and international cooperation between Japan, China and Korea: Present and future scenario analysis. *J. Environ. Sci.* 2011, 23, S162–S166. [CrossRef]
- 139. ABRELPE. Panorama dos Resíduos Sólidos no Brasil-2014 [Overview of Solid Waste in Brazil]. 2015. Available online: https://edisciplinas.usp.br/pluginfile.php/4389267/mod_resource/content/1/panorama2014.pdf (accessed on 18 July 2022).
- 140. Santini, A.; Morselli, L.; Passarini, F.; Vassura, I.; Di Carlo, S.; Bonino, F. End-of-Life Vehicles management: Italian material and energy recovery efficiency. *Waste Manag.* 2011, *31*, 489–494. [CrossRef]
- 141. Gallone, T.; Zeni-guido, A. Closed-Loop Opportunity for the Automotive. 2019. Available online: http://journals.openedition. org/factsreports/5225 (accessed on 18 July 2022).
- 142. Hjelmar, O. Disposal strategies for municipal solid waste incineration residues. J. Hazard. Mater. 1996, 47, 345–368. [CrossRef]
- 143. Das, S.; Curlee, T.R.; Rizy, C.G.; Schexnayder, S.M. Automobile recycling in the United States: Energy impacts and waste generation. *Resour. Conserv. Recycl.* **1995**, *14*, 265–284. [CrossRef]
- 144. Mancini, G.; Luciano, A.; Viotti, P.; Fino, D. Evaluation of automotive shredder residues (ASR) landfill behavior through lysimetric and traditional leaching tests. *Environ. Sci. Pollut. Res.* **2020**, *27*, 13360–13369. [CrossRef] [PubMed]
- 145. Cucchiella, F.; D'Adamo, I.; Gastaldi, M. Sustainable waste management: Waste to energy plant as an alternative to landfill. *Energy Convers. Manag.* 2017, 131, 18–31. [CrossRef]
- 146. Pan, S.Y.; Du, M.A.; Te Huang, I.; Liu, I.H.; Chang, E.E.; Chiang, P.C. Strategies on implementation of waste-to-energy (WTE) supply chain for circular economy system: A review. *J. Clean. Prod.* **2015**, *108*, 409–421. [CrossRef]
- 147. Šomplák, R.; Touš, M.; Pavlas, M.; Gregor, J.; Popela, P.; Rychtář, A. Multi-commodity network flow model applied to waste processing cost analysis for producers. *Chem. Eng. Trans.* **2015**, *45*, 733–738. [CrossRef]
- 148. Šomplák, R.; Pavlas, M.; Nevrlý, V.; Touš, M.; Popela, P. Contribution to Global Warming Potential by waste producers: Identification by reverse logistic modelling. *J. Clean. Prod.* **2019**, *208*, 1294–1303. [CrossRef]
- 149. De Assis, T.F.; de Abreu, V.H.S.; da Costa, M.G.; D'Agosto, M.d.A. Methodology for Prioritizing Best Practices Applied to the Sustainable Last Mile—The Case of a Brazilian Parcel Delivery Service Company. *Sustainability* **2022**, *14*, 3812. [CrossRef]